Author’s response to reviews

Title: Long-term follow-up in a Chinese child with congenital lipoid adrenal hyperplasia due to a StAR gene mutation

Authors:

Xiu Zhao (zoeman221@163.com)
Zhe Su (su_zhe@126.com)
Xia Liu (1136886046@qq.com)
Jianming Song (jamie898@163.com)
Yungen Gan (gdszgyg-l@126.com)
Pengqiang Wen (wenpq@live.cn)
Shoulin Li (lishoulinsz@126.com)
Li Wang (p.bear@yeah.net)
Lili Pan (panlilier@163.com)

Version: 3 Date: 11 Oct 2018

Author’s response to reviews:

Dear Editor:

Thank you for your kind letter titled “Your submission to BMC Endocrine Disorders - BEND-D-18-00178R2” on Oct 10, 2018. We revised the manuscript in accordance with the your comments. We had submitted a clean version of the manuscript.

We describe our revisions according to your comments below.

1. In the consent for publication please clarify if consent was given by the parent/legal guardian for publication of images.

The authors’ answer: Thank you very much for the thoughtful advice. The patient’s father had given us the written consent for publication of images. We have amended the sentences to clarify the situation in Consent for publication.
2. Please amend the abstract you have provided in the submissions system so that it matches the version in the manuscript.

The authors’ answer: Thank you for your advice. We have amended the abstract in the submissions system.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The authors’ answer: Thank you for the helpful advice. We have amended the manuscript and uploaded the manuscript according to your suggestions.

We sincerely appreciate your comments and suggestions, which have been valuable for improving the quality of our manuscript.

Thank you for the thoughtful advice.

Sincerely yours,

Xiu Zhao